Author’s response to reviews

Title: Molecular diagnosis in recessive pediatric neurogenetic disease can help reduce disease recurrence in families

Authors:

Mahmoud Issa (myissa2002@hotmail.com)
Zinayida Schlachetzki (z.schlachetzki@gmail.com)
Valentina Stanley (vstanley@ucsd.edu)
Renee George (reneegeorge@gmail.com)
Jennifer McEvoy-Venneri (jmcvoyvenneri@ucsd.edu)
Denice Belandres (dbelandres@ucsd.edu)
Hasnaa Elbendary (hasnaa_mohamed24@yahoo.com)
Khaled Gaber (krgaber@gmail.com)
Ahmed Nabil (ahmadnabil248@yahoo.com)
Mohamed Abdel-Hamid (mohamadnrc@hotmail.com)
Maha Zaki (dr_mahazaki@yahoo.com)
Joseph Gleeson (jogleeson@ucsd.edu)

Version: 5 Date: 21 Apr 2020

Author’s response to reviews:

Dear Editor,

We have addressed the issue with consent for publication that you identified. Your specific comment was:
"1. Consent for publicationThank you for your response. However, your statement “Written consent for publication of identifying images and other personal and clinical details of participants that compromise anonymity was documented in the consent form” is unclear, as it is unclear whether the consent form was signed by the participants. We kindly request that, in your “Consent for publication” section, you confirm that written informed consent for publication of personal details, clinical details and/or clinical images was obtained from all of the participants aged 18 and over."
We have amended the Consent for Publication statement to the following:
"Written consent for publication of identifying images and other personal and clinical details of participants that compromise anonymity was documented in the consent form that was signed by all participants. Written consent for publication of identifying images or other personal or clinical details of participants under the age of 18 that compromise anonymity was documented in the consent form and signed by parents or legal guardians."

Please let us know if there are any other changes required prior to acceptance.

Many thanks,

Prof. Joseph Gleeson